Author's response to reviews

Title: Lack of association between the mutations of the gene encoding mitochondrial D310 (displacement loop) mononucleotide repeat and oxidative stress in chronic dialysis patients in Taiwan

Authors:

Jin-Bor Chen (jbchen1@ms5.hinet.net)
Tsu-Kung Lin (tklin@adm.cgmh.org.tw)
Shang-Chih Liao (shangchi@adm.cgmh.org.tw)
Wen-Chin Lee (leewc@adm.cgmh.org.tw)
Lung-Chih Lee (longee@ms19.url.com.tw)
Chia-Wei Liou (cwliou@ms22.hinet.net)
Pei-Wen Wang (wangpw@ms18.hinet.net)
Mao-Meng Tiao (tmm@cgmh.org.tw)

Version: 3 Date: 2 November 2009

Author's response to reviews:

We had revised the changes of table and figure according to the instruction.